\documentclass[kind=GCE,a4j,bachelor,landscape,viewer,color,check]{edb-syl}
\def\�N�x{2011}
\def\�a�N�x{23}
\begin{document}

\Englishfalse
%%%% 0:220890
\begin{edb:lecture}%
\EdbEID{220890}\EdbEOID{557101}%
\���false%
\����ID{218270:220890}
\���ƊT�v���{166225}%
\����{�p��}{English}%
\���{���M�^�p��}{Communicative English}%
\�Ώ�{����}{(�БnE)2�N}%
\�J�u{�O}{��}{7�E8}%
\�S������{\������������{��w�@�\�V�I�E�A�[�c�E�A���h�E�T�C�G���X������}{�u�t}{\href{http://pub2.db.tokushima-u.ac.jp/ERD/person/152481/profile-ja.html}{\EdbSMG{�X�e�B�[�����Y}{�A��}{�����f�B�X}}}}%
\EdbColDef{edb:lecture.goal}{1}{E}{Students will develop confidence in oral communication and refine their listening skills.}%
\EdbColDef{edb:lecture.goal}{1}{J}{Students will develop confidence in oral communication and refine their listening skills.}%
\EdbColDef{edb:lecture.plan}{1}{E}{Geography and Climate}%
\EdbColDef{edb:lecture.plan}{1}{J}{Geography and Climate}%
\EdbColDef{edb:lecture.plan}{2}{E}{Homestays}%
\EdbColDef{edb:lecture.plan}{2}{J}{Homestays}%
\EdbColDef{edb:lecture.plan}{3}{E}{Family Roles}%
\EdbColDef{edb:lecture.plan}{3}{J}{Family Roles}%
\EdbColDef{edb:lecture.plan}{4}{E}{Transport}%
\EdbColDef{edb:lecture.plan}{4}{J}{Transport}%
\EdbColDef{edb:lecture.plan}{5}{E}{The Royal Show}%
\EdbColDef{edb:lecture.plan}{5}{J}{The Royal Show}%
\EdbColDef{edb:lecture.plan}{6}{E}{Language Schools and Universities}%
\EdbColDef{edb:lecture.plan}{6}{J}{Language Schools and Universities}%
\EdbColDef{edb:lecture.plan}{7}{E}{South Australia}%
\EdbColDef{edb:lecture.plan}{7}{J}{South Australia}%
\EdbColDef{edb:lecture.plan}{8}{E}{Sport}%
\EdbColDef{edb:lecture.plan}{8}{J}{Sport}%
\EdbColDef{edb:lecture.plan}{9}{E}{Australian Flora}%
\EdbColDef{edb:lecture.plan}{9}{J}{Australian Flora}%
\EdbColDef{edb:lecture.plan}{10}{E}{Australian Fauna}%
\EdbColDef{edb:lecture.plan}{10}{J}{Australian Fauna}%
\EdbColDef{edb:lecture.plan}{11}{E}{The Aborigines}%
\EdbColDef{edb:lecture.plan}{11}{J}{The Aborigines}%
\EdbColDef{edb:lecture.plan}{12}{E}{The Sea}%
\EdbColDef{edb:lecture.plan}{12}{J}{The Sea}%
\EdbColDef{edb:lecture.plan}{13}{E}{Mutliculturalism}%
\EdbColDef{edb:lecture.plan}{13}{J}{Mutliculturalism}%
\EdbColDef{edb:lecture.plan}{14}{E}{Australian English}%
\EdbColDef{edb:lecture.plan}{14}{J}{Australian English}%
\EdbColDef{edb:lecture.plan}{15}{E}{Test}%
\EdbColDef{edb:lecture.plan}{15}{J}{Test}%
\EdbColDef{edb:lecture.plan}{16}{E}{Review}%
\EdbColDef{edb:lecture.plan}{16}{J}{Review}%
\EdbColDef{edb:lecture.style}{1}{@}{101770}%
\EdbColDef{edb:lecture.style}{1}{E}{���̑�}%
\EdbColDef{edb:lecture.style}{1}{J}{���̑�}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.target}{1}{E}{To develop listening and speaking skills through purposeful interaction.}%
\EdbColDef{edb:lecture.target}{1}{J}{To develop listening and speaking skills through purposeful interaction.}%
\EdbColDef{edb:lecture.outline}{1}{E}{The class will begin with a dictation.\ We will then do communication activities and complete work from the textbook.}%
\EdbColDef{edb:lecture.outline}{1}{J}{The class will begin with a dictation.\ We will then do communication activities and complete work from the textbook.}%
\EdbColDef{edb:lecture.keyword}{1}{E}{Listening}%
\EdbColDef{edb:lecture.keyword}{1}{J}{Listening}%
\EdbColDef{edb:lecture.keyword}{2}{E}{Dictation}%
\EdbColDef{edb:lecture.keyword}{2}{J}{Dictation}%
\EdbColDef{edb:lecture.keyword}{3}{E}{Australian Culture}%
\EdbColDef{edb:lecture.keyword}{3}{J}{Australian Culture}%
\EdbColDef{edb:lecture.notice}{1}{E}{Punctuality and regular attendance are extremely important.\ Up to three absences will be permitted.\ Students are expected to communicate actively with all members of the class in English.}%
\EdbColDef{edb:lecture.notice}{1}{J}{Punctuality and regular attendance are extremely important.\ Up to three absences will be permitted.\ Students are expected to communicate actively with all members of the class in English.}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Speaking Tasks,{\EdbZSp} Homework,{\EdbZSp} Tests}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{Speaking Tasks,{\EdbZSp} Homework,{\EdbZSp} Tests}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{Only possible with a medical certificate or other official documentation.}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{Only possible with a medical certificate or other official documentation.}%
\EdbColDef{edb:lecture.textbook}{1}{E}{Modern Life Down Under,{\EdbZSp} Meredith Stephens and Richard Blight}%
\EdbColDef{edb:lecture.textbook}{1}{J}{Modern Life Down Under,{\EdbZSp} Meredith Stephens and Richard Blight}%
\EdbColDef{edb:lecture.contact}{1}{J}{\�A����l{\EdbEMAIL{meredith@ias.tokushima-u.ac.jp}{m{\HZSp[1]}e{\HZSp[1]}r{\HZSp[1]}e{\HZSp[1]}d{\HZSp[1]}i{\HZSp[1]}t{\HZSp[1]}h{\HZSp[1]}@{\HZSp[1]}i{\HZSp[1]}a{\HZSp[1]}s{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p}}{�X�e�B�[�����Y (3319, 098-656-7133, \EdbEMAIL{meredith@ias.tokushima-u.ac.jp}{m{\HZSp[1]}e{\HZSp[1]}r{\HZSp[1]}e{\HZSp[1]}d{\HZSp[1]}i{\HZSp[1]}t{\HZSp[1]}h{\HZSp[1]}@{\HZSp[1]}i{\HZSp[1]}a{\HZSp[1]}s{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p})}{}}%
\��������{�p��(2)}%
\������{���M�^�p��}%
\end{edb:lecture}%

\Englishtrue
%%%% 0:220890
\begin{edb:lecture}%
\EdbEID{220890}\EdbEOID{557101}%
\���false%
\����ID{218270:220890}
\���ƊT�v���{166225}%
\����{English}{�p��}%
\���{Communicative English}{���M�^�p��}%
\�Ώ�{����}{(�БnE)2�N}%
\�J�u{�O}{��}{7�E8}%
\�S������{\������������{Institute of Socio-Arts and Sciences}{Associate Professor}{\href{http://pub2.db.tokushima-u.ac.jp/ERD/person/152481/profile-en.html}{\EdbGMS{Meredith}{Anne}{Stephens}}}}%
\EdbColDef{edb:lecture.goal}{1}{J}{Students will develop confidence in oral communication and refine their listening skills.}%
\EdbColDef{edb:lecture.goal}{1}{E}{Students will develop confidence in oral communication and refine their listening skills.}%
\EdbColDef{edb:lecture.plan}{1}{J}{Geography and Climate}%
\EdbColDef{edb:lecture.plan}{1}{E}{Geography and Climate}%
\EdbColDef{edb:lecture.plan}{2}{J}{Homestays}%
\EdbColDef{edb:lecture.plan}{2}{E}{Homestays}%
\EdbColDef{edb:lecture.plan}{3}{J}{Family Roles}%
\EdbColDef{edb:lecture.plan}{3}{E}{Family Roles}%
\EdbColDef{edb:lecture.plan}{4}{J}{Transport}%
\EdbColDef{edb:lecture.plan}{4}{E}{Transport}%
\EdbColDef{edb:lecture.plan}{5}{J}{The Royal Show}%
\EdbColDef{edb:lecture.plan}{5}{E}{The Royal Show}%
\EdbColDef{edb:lecture.plan}{6}{J}{Language Schools and Universities}%
\EdbColDef{edb:lecture.plan}{6}{E}{Language Schools and Universities}%
\EdbColDef{edb:lecture.plan}{7}{J}{South Australia}%
\EdbColDef{edb:lecture.plan}{7}{E}{South Australia}%
\EdbColDef{edb:lecture.plan}{8}{J}{Sport}%
\EdbColDef{edb:lecture.plan}{8}{E}{Sport}%
\EdbColDef{edb:lecture.plan}{9}{J}{Australian Flora}%
\EdbColDef{edb:lecture.plan}{9}{E}{Australian Flora}%
\EdbColDef{edb:lecture.plan}{10}{J}{Australian Fauna}%
\EdbColDef{edb:lecture.plan}{10}{E}{Australian Fauna}%
\EdbColDef{edb:lecture.plan}{11}{J}{The Aborigines}%
\EdbColDef{edb:lecture.plan}{11}{E}{The Aborigines}%
\EdbColDef{edb:lecture.plan}{12}{J}{The Sea}%
\EdbColDef{edb:lecture.plan}{12}{E}{The Sea}%
\EdbColDef{edb:lecture.plan}{13}{J}{Mutliculturalism}%
\EdbColDef{edb:lecture.plan}{13}{E}{Mutliculturalism}%
\EdbColDef{edb:lecture.plan}{14}{J}{Australian English}%
\EdbColDef{edb:lecture.plan}{14}{E}{Australian English}%
\EdbColDef{edb:lecture.plan}{15}{J}{Test}%
\EdbColDef{edb:lecture.plan}{15}{E}{Test}%
\EdbColDef{edb:lecture.plan}{16}{J}{Review}%
\EdbColDef{edb:lecture.plan}{16}{E}{Review}%
\EdbColDef{edb:lecture.style}{1}{@}{101770}%
\EdbColDef{edb:lecture.style}{1}{J}{���̑�}%
\EdbColDef{edb:lecture.style}{1}{E}{���̑�}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.target}{1}{J}{To develop listening and speaking skills through purposeful interaction.}%
\EdbColDef{edb:lecture.target}{1}{E}{To develop listening and speaking skills through purposeful interaction.}%
\EdbColDef{edb:lecture.outline}{1}{J}{The class will begin with a dictation.\ We will then do communication activities and complete work from the textbook.}%
\EdbColDef{edb:lecture.outline}{1}{E}{The class will begin with a dictation.\ We will then do communication activities and complete work from the textbook.}%
\EdbColDef{edb:lecture.keyword}{1}{J}{Listening}%
\EdbColDef{edb:lecture.keyword}{1}{E}{Listening}%
\EdbColDef{edb:lecture.keyword}{2}{J}{Dictation}%
\EdbColDef{edb:lecture.keyword}{2}{E}{Dictation}%
\EdbColDef{edb:lecture.keyword}{3}{J}{Australian Culture}%
\EdbColDef{edb:lecture.keyword}{3}{E}{Australian Culture}%
\EdbColDef{edb:lecture.notice}{1}{J}{Punctuality and regular attendance are extremely important.\ Up to three absences will be permitted.\ Students are expected to communicate actively with all members of the class in English.}%
\EdbColDef{edb:lecture.notice}{1}{E}{Punctuality and regular attendance are extremely important.\ Up to three absences will be permitted.\ Students are expected to communicate actively with all members of the class in English.}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{Speaking Tasks,{\EdbZSp} Homework,{\EdbZSp} Tests}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Speaking Tasks,{\EdbZSp} Homework,{\EdbZSp} Tests}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{Only possible with a medical certificate or other official documentation.}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{Only possible with a medical certificate or other official documentation.}%
\EdbColDef{edb:lecture.textbook}{1}{J}{Modern Life Down Under,{\EdbZSp} Meredith Stephens and Richard Blight}%
\EdbColDef{edb:lecture.textbook}{1}{E}{Modern Life Down Under,{\EdbZSp} Meredith Stephens and Richard Blight}%
\EdbColDef{edb:lecture.contact}{1}{E}{\�A����l{\EdbEMAIL{meredith@ias.tokushima-u.ac.jp}{m{\HZSp[1]}e{\HZSp[1]}r{\HZSp[1]}e{\HZSp[1]}d{\HZSp[1]}i{\HZSp[1]}t{\HZSp[1]}h{\HZSp[1]}@{\HZSp[1]}i{\HZSp[1]}a{\HZSp[1]}s{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p}}{Stephens (3319, +81-98-656-7133, \EdbEMAIL{meredith@ias.tokushima-u.ac.jp}{m{\HZSp[1]}e{\HZSp[1]}r{\HZSp[1]}e{\HZSp[1]}d{\HZSp[1]}i{\HZSp[1]}t{\HZSp[1]}h{\HZSp[1]}@{\HZSp[1]}i{\HZSp[1]}a{\HZSp[1]}s{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p})}{}}%
\��������{�p��(2)}%
\������{���M�^�p��}%
\end{edb:lecture}%

\end{document}
